Author's response to reviews

Title: Efficacy of low level laser therapy on neurosensory recovery after injury to the inferior alveolar nerve

Authors:

- Tuncer Ozen (ozen@gata.edu.tr)
- Kaan Orhan (call53@yahoo.com)
- Ilker Gorur (ilkergorur@yahoo.com)
- Adnan Ozturk (ozturk@dentistry.ankara.edu.tr)

Version: 3 Date: 4 February 2006

Author's response to reviews: see over
Dear Editor;

First of all thank you very much again for your kind concern about our manuscript. Secondly, as you mentioned, we add a section in the acknowledgements part about Prof. Walsh kind help concerning the English revision and preparation of the manuscript. Also we add and changed author’s contributions part. Finally as you mentioned, we corrected and prepared the manuscript according to “pre- acceptance” checklist and resubmitting the changed manuscript via the submission system.

Once again thank you very much for your kind concern.

With my King Regards

Kaan Orhan

University of Ankara, Faculty of dentistry
Department of Oral Diagnosis and Radiology.

06500- Besevler-Ankara-TURKEY
Fax Number: 00 90 312 212 39 54,
Telephone Number: 00 90 312 212 62 50,
E-mail Address: call53@yahoo.com